Author's response to reviews

Title: Fusion of the BCL9 HD2 domain to E1A increases the cytopathic effect of an oncolytic adenovirus that targets colon cancer cells

Authors:

Christophe Fuerer (cfuerer@stanford.edu)
Krisztian Homicsko (Krisztian.Homicsko@isrec.unil.ch)
Alexander N Lukashev (Alexander.Lukashev@isrec.unil.ch)
Anne-Laure Pittet (Anne-Laure.Pittet@hospvd.ch)
Richard D Iggo (Richard.Iggo@isrec.unil.ch)

Version: 3 Date: 12 July 2006

Author's response to reviews:

I have added a methods section to the abstract, changed the order of the sections in the text, and added abbreviations, competing interests and authors' contributions sections.